Author’s response to reviews

Title: i-Scan detection of minimal change esophagitis in dyspeptic patients with or without Gastroesophageal Reflux disease.

Authors:
Nisa Netinatsunton (nisasan@yahoo.com)
Jaksin Sottisuporn (pondjaksin@hotmail.com)
Sririboon Attasaranya (sattasar@gmail.com)
Teepawit Witeerungrot (dr_pom@hotmail.com)
Naichaya Chamroonkul (naichaya@gmail.com)
Theeratus Jongboonyanuparp (theeratus@hotmail.com)
Alan Geater (alan.g@psu.ac.th)
Bancha Ovartlarnporn (mrovart@ji-net.com)

Version: 2 Date: 18 Nov 2015

Author’s response to reviews:

Dear editor,

We have revised our manuscript according to the suggestions.

1. We have added the trial registration information at the end of the abstract.

2. The statement of “All authors have read and approved the final version of the manuscript” was added.

3. We have added the email addresses and affiliations of all authors.

4. The legends of all the images were removed and put in the manuscript.